Author's response to reviews

Title: 47,XXY/48,XXXY/49,XXXXY mosaic with hydrocephaly: case report and literature review.

Authors:

Jesus E Duenas (jedanet@yahoo.com)
Maribel Aguilar (maribela2@excite.com)
Eliakym Arambula (eliakymarambula@hotmail.com)
Juliana B Valenzuela (juliana_70@yahoo.com)
Angelina Vega (angygfb@hotmail.com)
Julio Granados (julgrate@yahoo.com)
Rosalio Ramos (ramospayan@yahoo.com.mx)

Version: 2 Date: 29 May 2007

Author's response to reviews:

In this version, we delete figure 1 (photos of the patient), capitalize X and Y in the title and include the informed consent of the newborn's mother.